

# Individual response to lifestyle interventions: a pooled analysis of three long-term weight loss trials

Anat Yaskolka Meir (1)<sup>1,2†</sup>, Gal Tsaban (1)<sup>1,3†</sup>, Ehud Rinott (1)<sup>1</sup>, Hila Zelicha<sup>1</sup>, Dan Schwarzfuchs<sup>4</sup>, Yftach Gepner<sup>1,5</sup>, Assaf Rudich<sup>1</sup>, Ilan Shelef (1)<sup>4</sup>, Matthias Blüher<sup>6,7</sup>, Michael Stumvoll<sup>6,7</sup>, Uta Ceglarek (1)<sup>8</sup>, Berend Isermann (1)<sup>8</sup>, Nora Klöting<sup>6</sup>, Maria Keller<sup>6,7</sup>, Peter Kovacs (1)<sup>7</sup>, Lu Qi (1)<sup>9,10</sup>, Dong D. Wang<sup>10,11,12</sup>, Liming Liang<sup>2,13</sup>, Frank B. Hu<sup>2,10,11</sup>, Meir J. Stampfer<sup>2,10,11</sup>, and Iris Shai (1)<sup>1,7,10</sup>\*

<sup>1</sup>The International Center of Health Innovation & Nutrition in memory of Manya Igel, Faculty of Health Sciences, Ben-Gurion University of the Negev, P.O. Box 653, Beer-Sheva 8410501, Israel; <sup>2</sup>Department of Epidemiology, Harvard T.H. Chan School of Public Health, Boston, MA 02115, USA; <sup>3</sup>Department of Cardiovascular Medicine, Mayo Clinic, Rochester, MN 55905, USA; <sup>4</sup>Soroka University Medical Center, Beer-Sheva 8410501, Israel; <sup>5</sup>Department of Epidemiology and Preventive Medicine, School of Public Health, Sackler Faculty of Medicine, and Sylvan Adams Sports Institute, Tel-Aviv University, Tel-Aviv 6997801, Israel; <sup>6</sup>Helmholtz Institute for Metabolic, Obesity and Vascular Research (HI-MAG) of the Helmholtz Center Munich at the University of Leipzig and University Hospital Leipzig, Leipzig 04103, Germany; <sup>7</sup>Medical Department III—Endocrinology, Nephrology, Rheumatology, University of Leipzig Medical Center, Liebigstraße 20, Leipzig 04103, Germany; <sup>8</sup>Institute of Laboratory Medicine, Clinical Chemistry, and Molecular Diagnostics, University of Leipzig Medical Center, Leipzig 04103, Germany; <sup>9</sup>Department of Fublic Health and Tropical Medicine, Tulane University, New Orleans, LA 70118, USA; <sup>10</sup>Department of Nutrition, Harvard T.H. Chan School of Public Health, 655 Huntington Ave, Boston, MA 02115, USA; <sup>11</sup>Broad Institute of MIT and Harvard, Cambridge, MA 02142, USA; and <sup>13</sup>Department of Biostatistics, Harvard T.H. Chan School of Public Health, Boston, MA 02115, USA;

Received 11 January 2025; revised 16 April 2025; accepted 5 May 2025; online publish-ahead-of-print 5 June 2025

Aims	We explored the manifestations of individual weight loss (WL) response to long-term lifestyle interventions on cardiometabolic risk.
Methods and results	We pooled data from three large long-term lifestyle WL-intervention trials: 24-month DIRECT (ClinicalTrials.gov: NCT00160108; $n = 322$ ; 87% adherence), 18-month CENTRAL (ClinicalTrials.gov: NCT01530724; $n = 278$ ; 86% adherence), and 18-month DIRECT PLUS (ClinicalTrials.gov: NCT03020186; $n = 294$ ; 89% adherence). We analyzed longitudinal changes in cardiometabolic risk markers, including anthropometrics, blood biomarkers, and magnetic-resonance-imaging-assessed fat depots, and measured DNA-methylation, proteomics, and metabolomics. Among trial completers ( $n = 761$ , mean age = 50.4 years; 89% men, baseline body-mass-index = 30.1 kg/m <sup>2</sup> ), mean WL was $-3.3$ kg ( $-3.5\%$ ). We classified participants as Successful-WL (36%) with relative-WL > 5%, WL-Resistant (28%) who did not lose or gained weight, and Moderate-WL (36%) with WL between 0% and 5%. Successful-WL achieved the greatest improvements in multiple health indicators. However, the WL-Resistant also showed some significant improvements, with increased high-density-lipoprotein-cholesterol (HDLc) and decreased leptin and visceral fat ( $P < 0.05$ vs. baseline). Overall, each 1 kg sustained lifestyle-induced WL was associated with improvements in lipid markers and insulin resistance [HDLc (+1.44%), triglycerides ( $-1.37\%$ ), insulin ( $-2.46\%$ ), HOMA-IR ( $-2.71\%$ ), leptin ( $-2.79\%$ )] and intrahepatic-fat regression ( $-0.49$ absolute-units)] and modest but significant change in systolic and diastolic blood pressures ( $-0.26\%$ and $-0.36\%$ ). We identified 12 significant methylation sites that are associated with Successful-WL (FDR < 0.05; AUC = 0.73).
Conclusion	While only ~one-third of individuals achieved long-term WL, the Moderate-WL and WL-Resistant individuals could benefit improvements in visceral adiposity and cardiometabolic risk by shifting towards a healthy lifestyle pattern, beyond WL. Site-pecific DNA methylation may predict an individual's likelihood of successful WL.
Registration	NCT00160108, NCT01530724, NCT03020186.

<sup>\*</sup> Corresponding author. Tel: +972 8 647 7443, Email: irish@bgu.ac.il

 $<sup>^{\</sup>dagger}$  The first two authors contributed equally to the study.

<sup>©</sup> The Author(s) 2025. Published by Oxford University Press on behalf of the European Society of Cardiology. All rights reserved. For commercial re-use, please contact reprints@oup.com for reprints and translation rights for reprints. All other permissions can be obtained through our RightsLink service via the Permissions link on the article page on our site—for further information please contact journals.permissions@oup.com.

#### Lay summary

- In a pooled analysis of over 700 participants from three long-term lifestyle intervention trials, only one-third achieved significant WL (>5%). However, even individuals resistant to WL demonstrated meaningful health benefits.
  - These benefits included improvements in cardiometabolic markers such as increased HDL cholesterol, reduced visceral fat, and lower leptin levels, suggesting that lifestyle changes can enhance metabolic health independent of weight reduction.
  - Additionally, researchers identified specific DNA methylation patterns that may predict an individual's likelihood of achieving successful WL through lifestyle modification.

#### **Graphical Abstract**



## Introduction

Weight loss (WL) can promote metabolic and cardiovascular health in a variety of ways, as reducing the risk of type 2 diabetes,<sup>1</sup> heart disease,<sup>2</sup> and hypertension<sup>3</sup> and reducing the risk of all-cause mortality for intentional WL.<sup>4</sup> WL therapy may be approached in a more personally tailored manner, adapted to the individual's genetic<sup>5,6</sup> and non-genetic characteristics.<sup>7–10</sup> A WL of at least 5% was associated with improved cardiometabolic risk factors, including total body fat, truncal fat, waist circumference (WC), systolic and diastolic blood pressure (BP), and high-density lipoprotein cholesterol (HDLc).<sup>11–14</sup>

Weight regain after WL is a common challenge; a meta-analysis of 29 directed WL studies showed a 25%, 50%, and 75% weight regain after 1, 2, and 5 years, respectively.<sup>15</sup> There is wide variation in how individuals respond to WL interventions of diet and exercise habits changes. Some individuals may see significant WL with these interventions, while others may experience minimal or no changes.<sup>16</sup> This variability may be due to various factors, including genetic,<sup>17</sup> metabolic,<sup>18</sup> and behavioural adaptations.<sup>19</sup> Initial WL during the first months of a diet, the 'rapid WL phase', is often the most substantial and is followed by weight regain or

maintenance ('plateau').<sup>20</sup> Factors contributing to weight regain include a slowing of metabolism,<sup>21</sup> changes in hormone levels<sup>22</sup> that stimulate appetite, and a return to previous dietary and lifestyle habits.

We aimed to estimate the extent of improvement in cardiometabolic outcomes per modest WL, further explored individual responses across categories of responders to lifestyle interventions for WL, and explored potential baseline indicators of successful WL. We pooled data from three landmark large long-term lifestyle WL-intervention trials: The DIRECT (n = 322; 24 months, 87% adherence-rate)<sup>23</sup> CENTRAL (n = 278; 18 months, 86% adherence-rate),<sup>24</sup> and DIRECT PLUS (n = 294; 18 months, 89% adherence-rate).<sup>25</sup>

# **Methods**

#### Study population

Individuals from three long-term, large-scale randomized controlled trials who completed the intervention were included in this analysis: DIRECT (NCT00160108; n completers = 272), CENTRAL (NCT01530724; n completers = 232), DIRECT PLUS (NCT03020186; n completers = 264). We

also performed a sub-analysis of abdominal and liver imaging data for the CENTRAL and DIRECT PLUS. Inclusion and exclusion criteria for each trial are detailed in Supplementary material online, Methods *S1*. Assessment of adherence to the intervention is detailed in Supplementary material online, Methods *S2*. Study protocols were published elsewhere.<sup>23–25</sup>

The DIRECT trial included three diet-only hypocaloric intervention arms: a low-fat diet, a low-carbohydrate diet, and a Mediterranean diet. The CENTRAL trial employed a two-factorial lifestyle intervention consisting of hypocaloric diets (similar in composition and caloric restriction to those in the DIRECT trial) followed by a lifestyle accommodation phase after 6 months. The diets in CENTRAL included a low-fat diet and a combined lowcarbohydrate/Mediterranean diet. The DIRECT PLUS trial included a combined diet and lifestyle intervention from the outset, comprising three regimens: a guideline-recommended healthy lifestyle without caloric restriction, a low-carbohydrate Mediterranean diet (similar to CENTRAL), and a green-Mediterranean diet-an adaptation of the low-carbohydrate Mediterranean diet with further red meat restriction and the addition of green tea and a Mankai shake as a dinner substitute. Despite some differences in intervention design, all three studies demonstrated similar patterns and magnitudes of WL: a peak after six months followed by a plateau or partial weight regain. Across all trials, WL emerged as the primary driver of cardiometabolic improvements, with modest but meaningful contributions from the specific lifestyle approaches.

#### WL response patterns

We used long-term WL to define retrospectively three categories of weight response by the end of intervention: Successful-WL: individuals who lost more than 5% of their initial body weight; Moderate-WL: individuals who lost 0–5%; WL-Resistant: individuals who completed the intervention but did not lose any weight or gained weight.

#### Measurement of outcome indicators

We evaluated weight, height, WC, systolic and diastolic BP, and blood biomarkers three times during the interventions, as detailed before.<sup>23–25</sup> Briefly, the participants were weighed without shoes to the nearest 0.1 kg. A wall-mounted stadiometer was used to measure height to the nearest millimetre. WC was measured halfway between the last rib and the iliac crest. BP was measured using an automated system (Datascop Acutor 4). Blood samples were obtained by venipuncture at 8 a.m. after a 12-h fast at baseline, 6, and the trial end and were stored at –80°C until an assay for lipids, inflammatory biomarkers, and insulin could be performed at Leipzig University, Germany. For the CENTRAL and DIRECT PLUS trials, we assessed visceral adipose tissue (VAT) and intrahepatic fat percentage (IHF%) pre- and post-intervention (two timepoints) using magnetic resonance imaging (MRI), as detailed before.<sup>24,25</sup>

## **Omics sub-study to predict WL categories**

We used DNA methylation (Illumina EPIC array), proteomics (Olink CVD II), and plasma metabolomics (liquid chromatography-tandem mass spectrometry) data measured at baseline in the DIRECT PLUS and CENTRAL trials. Preprocessing information is available in Supplementary Methods S3. We employed two methods for the discovery of omics predictors for WL: (i) a wide association study: we used the DIRECT PLUS as a discovery set to predict WL in kg and the CENTRAL as a validation set to predict WL in kg, WL above/below 5%, and WL-resistant vs. Successful-WL. Models are detailed in Supplementary material online, Methods S4. (ii) An elastic net regression with Leave One Out cross-validation (LOOCV) to predict WL in kg by each omics in the DIRECT PLUS. The CENTRAL study was used to validate the prediction models. Models are detailed in Supplementary material online, Methods S4.

#### Statistical analysis

We primarily aimed to examine the association of 1 kg WL with multiple health outcomes. The initial sample size for this analysis was determined

by the number of people with baseline and long-term follow-up weight measurements (n = 768). For eligible observations, we removed outliers three standard deviations (SD) from the mean weight for baseline and end of the intervention (18 or 24 months) separately and then re-calculated the number of people with weight measurements at both time points (see Supplementary material online, Figure S1). Relative weight change was determined as (weight at the end of the intervention—weight at the beginning of the intervention)/weight at the beginning of the intervention × 100. Next, as needed, we removed values 3SD from the mean for the following markers for the 761 people mentioned above (see Supplementary material online, Methods S5). We calculated the difference between the end of the intervention (18 or 24 months; 'long-term WL') to the beginning using the clean data. We also calculated relative change (difference/baseline) × 100. We followed the same steps for our MRI sub-study, only within the trials with available data (CENTRAL, DIRECT PLUS). The P-value threshold for the primary outcome was corrected by dividing 0.05 by the number of tests. Continuous variables are presented as mean  $\pm$  SD. Nominal variables are expressed as numbers or percentages. Summary statistics were examined using the chi-square or ANOVA test. Benjamini Hochberg False Discovery Rate (FDR) was used to correct for multiple comparisons in the omics sub-study. Differences between time points (baseline and end of the intervention) were tested using the Paired sample t-test. A meta-analysis was conducted using the 'meta' R package. Interactions were tested using linear models, and P of interactions was reported for the interaction term. Multinomial multivariable regression was used to test differences between WL categories. To quantify the effect of WL on the percentage change in outcome indicators, we tested the assumption of whether linear models were correctly specified using the Ramsey Regression Equation Specification Error Test (RESET), 'Imtest' R package. When RESET test results were insignificant, we used Linear regression. Otherwise, we used cubic spline regression ('splines' R package). We performed a sensitivity analysis using Quintile regression using the 'quantreg' R package. All statistical analyses were performed using R (version 4.2; R Foundation for Statistical Computing).

# Results

## Characterization of WL categories by baseline characteristics

Baseline characteristics across three patterns of long-term WL categories are presented in *Table 1*. In a univariate test, the following markers significantly differ between the WL-Resistant and the other WL categories: younger age, higher percentage of females, lower WC (compared with Successful-WL), lower liver enzymes, and least IHF (least %). A meta-analysis for the potential differentiating factors observed in the univariate test (type of diet, sex, age, WC, leptin, fetuin-A, ALT, AST, VAT%, and IHF) between the extreme WL phenotypes WL-Resistant and Successful-WL yielded similar results (see Supplementary material online, *Figure S2*). After adjustment for age, sex, trial, and BMI, most of the characteristics were not significantly different between groups. Supplementary material online, *Table S1* presents a sensitivity multivariable analysis for markers that significantly differed between WL categories of below/above 5% at baseline. A sensitivity analysis of each trial separately showed similar results (see Supplementary material online, *Table S2*).

## Baseline omics as predictors of long-term WL and WL categories

We performed an exploratory analysis in a subset of individuals with DNA methylation, proteomics, and blood metabolomics to find predictors for WL, using Elastic net models with each pre-intervention omics as predictors for WL (in kg) and wide association study with each

	WL-Resistant (no WL; <i>n</i> = 212)	Moderate-WL (0–5% WL; <i>n</i> = 277)	Successful-WL (>5% WL; n = 272)					
Packground characteristics								
	49.8 (0.02)	F1 0 (0 40)						
Age, years	48.8 (9.03)	51.0 (8.48)	51.1 (9.35)					
Sex, % of men	84.4%	89.2%	91.9%					
Smokers, %	14.62%	16.77%	13.65%					
Oral glycemic control, %	5.66%	4.33%	6.25%					
Lipid-lowering, %	14.62%	15.88%	18.01%					
Antihypertensive, %	12.74%	19.86%	20.59%					
Anthropometric and BP								
BMI, kg/m <sup>2</sup>	30.9 (3.59)	30.3 (3.09)	30.8 (3.25)					
WC, cm	107 (9.66)	106 (8.52)	108 (8.43)					
Systolic BP, mmHg	127 (14.5)	129 (14.1)	129 (14.6)					
Diastolic BP, mmHg	79.4 (9.39)	80.5 (9.28)	79.7 (9.85)					
Blood markers								
HDLc, mg/dL	43.0 (10.7)	41.9 (10.4)	41.8 (9.89)					
LDLc, mg/dL	125 (27.2)	123 (32.1)	120 (31.6)					
Total cholesterol, mg/dL	197 (32.0)	196 (34.6)	193 (34.6)					
Triglycerides, mg/dL	152 (70.4)	157 (68.6)	155 (63.7)					
Fasting plasma glucose, mg/dL	98.1 (16.9)	97.7 (19.2)	97.2 (18.8)					
HOMA IR	3.64 (1.96)	3.52 (2.05)	3.42 (1.98)					
Insulin, µU/mL	14.6 (7.05)	13.9 (6.47)	14.2 (7.11)					
CRP, mg/L	3.10 (2.11)	2.99 (2.17)	3.11 (2.04)					
Leptin, ng/mL	13.0 (8.40)	11.2 (7.87) <sup>a</sup>	11.9 (7.73)					
Chemerin, ng/mL	199 (35.6)	198 (33.7)	197 (31.6)					
Fetuin-A, µg/mL	329 (94.1)	341 (83.3)	349 (85.6) <sup>b</sup>					
ALKP, U/L	70.8 (16.9)	72.8 (17.8)	72.1 (17.5)					
ALT, U/L	28.5 (13.4)	31.5 (13.6) <sup>b</sup>	31.9 (13.5) <sup>b</sup>					
AST, U/L	24.7 (7.55)	26.4 (8.13) <sup>a</sup>	26.7 (7.82)					
MRI-assessed major fat deposits	( ),		( )					
VAT. cm <sup>2</sup>	145 (61.0)	152 (53.5)	157 (56.3)					
VAT %	29.7 (10.0)	32.2 (8.93)	31.8 (9.71)					
Intrahepatic fat. %	8.44 (8.44)	10.9 (9.59) <sup>b</sup>	10.2 (9.10) <sup>a</sup>					
Retrospective intervention assignment			- \ - /					
MED and/or LC (vs. control), %	55.19%	58.48%	68.38% <sup>a,b</sup>					

#### Table 1 Baseline characteristics across three categories of long-term WL

<sup>a</sup>P-value < 0.1 in a multinomial regression with age, sex, trial, and baseline BMI as covariates; Moderate-WL as reference (highlighted in gray: P-value < 0.05). WL, weight loss. <sup>b</sup>P-value < 0.1 in a multinomial regression with age, sex, trial, and baseline BMI as covariates; WL-Resistant as reference (highlighted in gray: P-value < 0.05).

ALKP, Alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; BMI, body mass index; BP, blood pressure; CRP, C-reactive protein; HDLc, high-density lipoprotein cholesterol; HOMA IR, Homeostatic model assessment for insulin resistance; LC, low carbohydrates; LDL, low-density lipoprotein cholesterol; MED, Mediterranean; WL, weight loss.

pre-intervention omics associated with WL in kg. Elastic net models using DNA methylation, metabolomics, or proteomics as predictors did not perform well in predicting WL (see Supplementary material online, *Table S3*). In an epigenome-wide association study (by likelihood ratio test), we found 12 CpGs significantly associated with WL (FDR < 0.05) in our discovery set (DIRECT PLUS; *Figure 1A*; Supplementary material online, *Figure S4 and Table S4*). Using these CpGs in a validation set (CENTRAL) to predict WL below/above 5% in a logistic regression model yielded an AUC of 0.733 [95% CI (0.641, 0.825); *Figure 1D*]. Employing the 12 CpGs detected in the DIRECT PLUS epigenome-wide association study for WL in a cross-validated linear regression in the CENTRAL to predict WL (in kg) resulted in poor prediction

( $R^2 = 0.001$ , RMSE = 5.86), similar to the results of the elastic net with forced covariates ( $R^2 = 0.003$ , RMSE = 5.63; Supplementary material online, *Table S3*). We further examined the associations of proteomics and metabolomics with WL in our discovery set using multivariable linear models in the DIRECT PLUS (*Figure 1B* and *C*); No FDR < 0.05 markers were observed. Nevertheless, due to the exploratory nature of this analysis, we further used proteins and metabolites with P < 0.1 to predict WL above/below 5% in our validation set (CENTRAL). This yielded AUC of 0.611 [95% CI (0.534, 0.687)] and 0.609 [95% CI (0.530, 0.688)] proteomics and metabolomics, respectively (*Figure 1E* and *F*). Of note, serine, a metabolite found as a significant predictor in the training set, was not available in the CENTRAL data.



**Figure 1** Wide association studies for WL. An exploratory analysis. (A) Discovery, DNA methylation; (B) Discovery, proteomics; (C) Discovery, metabolomics; (D) Prediction of WL above/below 5% by selected CpGs; (E) Prediction of WL above/below 5% by selected proteins; Prediction of WL above/below 5% by selected metabolites. Red dashed line indicates FDR of 0.05. Blue dashed line indicates *P*-value of 0.1. Volcano plots: light blue dots indicates negative expression, and red dots indicate positive expression.

Using the proteins from DIRECT PLUS discovery set on the CENTRAL in a linear model without cross-validation resulted in  $R^2 = 0.019$  and RMSE = 5.56, similar to the LOOCV elastic net with forced covariate (see Supplementary material online, *Table S3*), compared with a cross-validated linear model with  $R^2 = 0.001$  and RMSE = 5.61. Replacing the

proteins with metabolites discovered in the DIRECT PLUS, linear models using the CENTRAL data in a cross-validated vs. no validation resulted in  $R^2 = 0.002$  vs.  $R^2 = 0.06$ .

We repeated the validation analysis using logistic regression comparing extreme categories of WL for prediction of Successful-WL vs.

				-		
Parameter: Baseline vs. end	WL-resistant $n = 212$		Moderate WL $n = 277$		Successful-WL $n = 272$	
	Mean	<i>P</i> -value <sup>a,b</sup>	Mean	<i>P</i> -value <sup>a,b</sup>	Mean	P-value <sup>a,b</sup>
Anthropometric and BP						
WC, cm	-0.32	0.34	-3.61	<0.001	-8.82	<0.001
WC (men only), cm	-0.62	0.08	-3.56	<0.001	-8.77	<0.001
WC (women only), cm	+1.37	0.21	-4.08	0.001	-9.38	<0.001
Systolic BP, mmHg	+1.19	0.14	-2.13	0.006	-3.89	<0.001
Diastolic BP, mmHg	+0.53	0.36	-0.99	0.0504	-2.97	<0.001
Blood markers						
HDLc, mg/dL	+1.16	0.008	+3.57	<0.001	7.64	<0.001
LDLc, mg/dL	+1.10	0.51	0.01	0.99	-0.72	0.68
Total cholesterol, mg/dL	+4.71	0.02	0.99	0.59	-0.96	0.61
Triglycerides, mg/dL	+6.42	0.08	-14.9	<0.001	-34.8	<0.001
Fasting plasma glucose, mg/dL	+2.34	0.003	+1.12	0.13	-1.05	0.16
HOMA IR	-0.02	0.86	-0.36	<0.001	-1.02	<0.001
Insulin, μU/mL	-0.30	0.40	-1.53	<0.001	-4.09	<0.001
CRP, mg/L	+0.23	0.08	-0.24	0.04	-0.82	<0.001
Leptin, ng/mL	-0.81	0.02	-1.82	<0.001	-5.15	<0.001
Chemerin, ng/mL	+6.28	0.02	-0.45	0.84	-6.54	0.002
Fetuin-A, μg/mL	-43.5	<0.001	-49.6	<0.001	-56.6	<0.001
ALKP, U/L	+0.46	0.49	-1.57	0.018	-2.27	<0.001
ALT, U/L	+2.65	0.002	-3.67	<0.001	-7.26	<0.001
AST, U/L	+0.46	0.42	-1.89	0.001	-2.79	<0.001
MRI-assessed major fat deposits						
VAT, cm <sup>2c</sup>	-7.15	<0.001	-22.9	<0.001	-56.9	<0.001
Intrahepatic fat, % <sup>c</sup>	-0.005	0.99	-3.11	<0.001	-7.11	<0.001

Table 2 Long-term within-group changes across three weight response categories

<sup>a</sup>Paired *T*-test.

<sup>b</sup>Bold values denote statistical significance at the P-value < 0.05 level; Italic values denote statistical significance at the P-value < 0.1 level.

<sup>c</sup>CENTRAL and DIRECT PLUS only.

ALKP, Alkaline phosphatase; ALT, alanine aminotransferase; AST, aspartate aminotransferase; BP, blood pressure; CRP, C-reactive protein; HDLc, high-density lipoprotein cholesterol; HOMA IR, Homeostatic model assessment for insulin resistance; LDL, low-density lipoprotein cholesterol; VAT, visceral adipose tissue; WC, waist circumference; WL, weight loss.

WL-resistant (see Supplementary material online, *Figure S4*). This slightly, but not significantly, improved the prediction by DNA methylation to an AUC of 0.791 [95% CI (0.684, 0.898)] and prediction by metabolomics AUC of 0.619 [95% CI (0.528, 0.701)]. The prediction by proteomics was not improved with an AUC of 0.604 [95% CI (0.509, 0.698)].

## Long-term changes within WL categories

We examined the long-term (*Table 2*) differences pre- and postintervention across the WL categories, as shown in *Figure 2A and B*. WL-resistant participants experienced HDLc elevation and MRI-assessed decrease of visceral and intra-hepatic fat (P < 0.05 for all, compared with baseline). However, there were increases in total cholesterol, chemerin, FPG, ALT, and a statistically marginal increase in triglycerides and CRP. Participants who reduced up to 5% of their initial body weight (Moderate-WL) improved the following health indicators: WC, systolic BP, HDLc, triglycerides, insulin, HOMA IR, CRP, leptin, fetuin-A, liver enzymes, abdominal fat depots areas, and liver fat. As expected, participants who lost more than 5% of their body weight (WL-Successors) showed the most improvement in health indicators, with all health indicators, excluding LDLc, total cholesterol, and FGP (P < 0.05 for all, baseline vs. T18). Key differences in health improvement between WL categories are summarized in Supplementary material online, *Figure S5*.

# The effect of moderate WL on long-term health indicators

For individuals who lost at least 1 kg (n = 482, 9.54% women) sustained for the long-term, we quantified the beneficial effect of every 1 kg WL on different health biomarkers. We tested the linearity between weight change (in kg) and each outcome as the relative change in the marker (3a presents selected pairs). For the pairs with RESET test P-value > 0.05, suggesting that a linear model is sufficient to explain the relationship between the dependent and independent variable, we present beta from the linear model, representing the magnitude of change for marker per 1-unit weight change (1 kg WL, Figure 3B, Supplementary material online, Table S5). After accounting for multiple linear testing, we observed the following change per -1 kg WL: HDLc (+1.44%), triglycerides (-1.37%), insulin (-2.46%), HOMA IR (-2.71%), leptin (-2.79%), and IHF (-0.49%). A modest but significant change was observed for chemerin (-0.47%), ALT (-0.73%), systolic and diastolic BPs (-0.26% and -0.36%). We repeated this analysis by adjusting for trial and examining WL×trial interactions (see Supplementary material online, Table S6). The interaction was only observed for ALKP (P of



Figure 2 Individual long-term changes. (A) WL (in %) in all three trials ordered from top WL-Resistant to top Successful-WL, with below WC and HDLc order by the WL extent, from top WL-Resistant to top Successful-WL. (B) WL (in %) in CENTRAL and DIRECT PLUS ordered from top WL-Resistant to top Successful-WL, with below VAT and IHF by WL extent, from top WL-Resistant to top Successful-WL.

interaction = 0.035). When adjusted for age, sex, and trial, the following outcomes improved by WL: all glycemic markers, some blood lipids, serum chemerin, leptin, diastolic and systolic BP, and IHF (all % change; Supplementary material online, *Table S7*). We did not observe any WL×sex interactions (see Supplementary material online, *Table S8*).

For CRP and VAT, we fitted cubic spline with the following cut points (see Supplementary material online, *Figure S6*): -1 kg, -2 kg, -3 kg, -4 kg, -5 kg, -10 kg, and -15 kg. Although there were significant knots, CRP did not show any obvious pattern. For VAT, significant knots were observed with increased WL.

As a sensitivity analysis, we performed quantile regression with 10th-90th percentiles for the markers showing significant associations after accounting for multiple testing (see Supplementary material online, *Table S9*; Scatter plots of these markers are available in

Supplementary material online, *Figure* S7). This analysis yielded results similar to those of linear regression.

# Discussion

This study pooled patient-level data from three long-term WL-intervention trials, suggesting that while only one-third achieve long-term Successful-WL, WL-Resistant individuals may benefit from lifestyle interventions even without WL. Each 1-kg sustained intentional lifestyle-induced WL was associated with significant improvement in cardiometabolic risk and atherosclerosis and insulin resistance promoting ectopic fats. Specific omics signatures, such as DNA-methylation, were related to WL individual response. Our findings represent an



**Figure 3** Linearity and the association between weight change (kg) and markers change (%). (A) Plots of a fitted model, smoothing spline for weight change (kg) and each marker change (%)—selected markers. *P*-value for RESET test presented for detection of mis-specified linear models. ss = smoothing spline line. Im = linear model line. (B) Beta coefficients for linear models. \* Denotes significance of predictor while adjusting for multiple testing (N = 16 linear tests) at the level of *P*-value < 0.003. The analysis included individuals who lost at least 1 kg (N = 482).

'average predicted response' across all participants, irrespective of the specific lifestyle intervention. This aligns with our study's main goal: characterize the generalisable association between WL and cardiometabolic improvements during structured lifestyle interventions.

Our analysis has some limitations. The WL-intervention trials included in this study had different follow-up times of 18 months (CENTRAL and DIRECT PLUS) and 24 months (DIRECT), although the inclusion and exclusion criteria for all studies were similar, and the population participating in all studies was from the same workplace and geographical location. Second, we did not stratify our analyses per individual adherence or specific intervention. Nevertheless, based on our previous reports, most of the participants highly adhered to their assigned intervention for the entire intervention period, as reported by self-reports (DIRECT, CENTAL, DIRECT PLUS)^{23-25} or estimated by objective measurements (DIRECT PLUS).<sup>25</sup> Moreover, all the intervention arms were active at the same intensity, healthy, and with no passive control groups. Third, the small number of women participating in these cohorts did not allow us to quantify the effect of 1 kg WL using subgroup analysis in women. The high proportion of men participants reflects the workplace profile, as acknowledged before.<sup>25</sup> This limits our ability to extrapolate our results to women. Due to a small sample size, we were underpowered to detect associations in some of the omics studies. Additionally, our elastic net models performed poorly when predicting WL using either omics. Elastic net models' lack of predictive performance may reflect potential overfitting during model training. Although some CpGs were significantly associated with the outcome in the wide association study, their effect sizes were small

and insufficiently robust to generalize across cross-validation folds. Given the relatively small sample size and the high-dimensional nature of the omics data, the models may have fit to noise or spurious patterns in the training set that did not hold in the validation folds. This is a wellknown risk in high-dimensional settings, even when using regularized models such as elastic net, particularly when the true signal is weak or diffuse. However, the omics sub-study is exploratory and should be used as preliminary data for future larger-scaled omics studies. A key strength of our study is using three intensive trials with high adherence. Another strength of our study lies in the rigorous and multifaceted approach used to evaluate the robustness of our findings. We conducted a series of sensitivity analyses to assess the stability of results under varying model specifications and assumptions, thereby reducing the likelihood that analytical choices drove findings. Stratified analyses by trial further allowed us to explore potential effect modification by key demographic or clinical variables, offering insights into populationspecific associations and enhancing interpretability. Additionally, we performed a meta-analysis across trials, explicitly modelling betweenstudy heterogeneity. This validated the pooled findings and ensured that any single trial did not disproportionately influence the observed differences between WL categories. We conducted sensitivity analyses using alternative modelling strategies, including replacing linear models with quantile regression and multiple multivariable models, to assess whether associations held across the outcome distribution rather than relying solely on mean effects. Additionally, we examined results across different subgroups (three WL categories and below/above 5% WL) to evaluate the consistency and potential effect modification

by key characteristics. Finally, we explored interactions between WL and both sex and cohort, allowing us to assess potential effect modification and uncover population-specific differences.

Our primary aim was to quantify sustainable WL's effect in cardiometabolic blood indicators of lipids, glycemic, liver enzymes, inflammation, and adipokines. Previous studies have assessed the beneficial effect of each 1 kg of WL on different health outcomes. A meta-analysis of 30 randomized controlled trials of diet and/or exercise intervention with follow-up of at least six months guantified the beneficial effect of WL on blood lipids in 2434 adults.<sup>26</sup> After 12 months, for every 1 kg of WL, triglycerides were reduced by -4.0 mg/dL, total cholesterol by -1.66 mg/dL, LDLc by -1.28 mg/dL, and HDLc increased by 0.46 mg/dL. Another meta-analysis of 43 studies with a median followup time of 6 months found that every 1 kg WL led to 0.83 U/L of ALT, 0.56 U/L of AST, and a 0.77% point reduction in hepatosteatosis.<sup>27</sup> Our current analysis confirms the effect of 1 kg WL for 18- or 24-month well-controlled lifestyle intervention on blood lipids and liver enzymes. It further suggests information on improving other blood biomarkers, including glycemic indicators and abdominal and intrahepatic fats.

Our secondary aim was to profile different categories of WL: successful, moderate, and no WL groups. A previous study of unsuccessful responders to a low-energy diet<sup>16</sup> reported that unsuccessful responders had higher resting heart rates, self-reported perceived stress, higher dietary restraint levels, and lower disinhibition levels than successful responders. In our current study, WL-Resistant had a lower percentage of individuals assigned to Mediterranean and/or Low carbohydrate diets (vs. control), were more likely to be women, younger, with lower baseline weight and WC, liver enzymes, and less IHF. This group also tended to have lower VAT% and fetuin-A and higher leptin. As shown in previous studies, higher weight reductions led to more remarkable improvement in cardiometabolic health indicators,<sup>11–14</sup> including ectopic fat depots. Yet, in our study, individuals who did not lose weight by the end of 18 and 24 months of the intervention improved their health profile, with significant reductions in leptin, abdominal adiposity, WC (in men), and an increase in HDLc, though their adverse changes in total cholesterol, FPG, and ALT levels.

In a previous epigenetic analysis of the CENTRAL trial, we observed that WL responders (top 10% of WL in the cohort) had differential DNA methylation at specific genomic regions, compared with non-responders (bottom 10% of WL in the cohort).<sup>28</sup> Our current results highlighting selected CpGs as predictors for WL (kg) are exploratory and should be interpreted with caution. The previous and current findings raise the possibility that successful or unsuccessful WL may be pre-identified based on various markers, including biological molecules.

The long-term sustainability of lifestyle-induced changes in cardiometabolic health beyond the trial period was demonstrated in several studies that performed follow-up on individuals who enrolled in intervention studies. The 4-year follow-up of the 2-year DIRECT trial showed favourable post-intervention effects on blood lipids, particularly among participants receiving the Mediterranean and lowcarbohydrate diets, despite a partial regain of weight.<sup>29</sup> A five-year follow-up of a randomized WL trial on a digital health behaviour change support system (HBCSS) showed no differences in maintaining reduced weight between the intervention groups (a web-based HBCSS utilising persuasive systems design and cognitive behavioural therapy methods).<sup>30</sup> However, individuals in the HBCSS group had a decrease in the need for antihypertensives at the 5-year checkpoint compared with the other groups. In the Look AHEAD lifestyle intervention trial among overweight/obese individuals with type 2 diabetes, larger WL during the intervention period produced greater improvements in

glycemic markers, blood lipids, and systolic BP at years 1 and 4.<sup>31</sup> Despite maintaining WL, HbA1c levels worsened between years 1 and 4 and remained below baseline only in those with large WL. Importantly, individuals who had large initial WL but full regain of weight had greater improvements in HbA1c levels at year 4 than those with smaller or no initial WL. These findings suggest that the magnitude of initial WL may have lasting metabolic benefits, even without sustained weight reduction.

Lifestyle interventions are the backbone of cardiovascular prevention. Obesity is a chronic disease, largely driven by long-term lifestyle choices, and is now increasingly recognized as a modifiable risk factor for cardiovascular disease.<sup>32</sup> Given that lifestyle-induced WL is often modest yet frequently accompanied by meaningful cardiometabolic benefits, the current study sought to quantify and characterize the expected improvements in cardiometabolic health parameters per unit of WL, independent of the intervention method. We believe our findings hold significant clinical relevance. In routine practice, healthcare providers are often asked to advise patients on the expected benefits of lifestyle interventions. While such projections are well established for certain pharmacologic therapies, the anticipated cardiometabolic effects of lifestyle-induced WL remain less clearly defined, particularly those not tied to a specific dietary regimen, remain less clearly defined.<sup>33</sup> Our study addresses this important gap by providing objective, evidence-based estimates that can support clinical decision-making and enhance patient counselling. As the management of obesity gains increasing prominence in cardiovascular prevention, these findings underscore the critical value of adopting a healthy lifestyle and the measurable benefits it can yield.

# Future perspective and conclusions

Cumulative exposure to unfavourable cardiometabolic status is linked to higher risks of type 2 diabetes, atherosclerosis progression, heart failure, and mortality<sup>34–36</sup>. Cardiometabolic prevention is now at the forefront of preventive cardiology efforts, with strong evidence emphasising the importance of diagnosing and treating obesity.<sup>37–39</sup> However, with the growing reliance on pharmaceutical and surgical interventions, the critical role of lifestyle management in cardiometabolic health is often overlooked. Additionally, dedicated studies focused on women are necessary to better characterize the relationship between lifestyle-induced WL and cardiometabolic health in this understudied population. Our study offers new insights into the cardiometabolic response to WL in individuals with obesity, highlighting the crucial, WL-independent benefits of maintaining a healthy lifestyle for reducing cardiovascular risk. Our results suggest the benefits of switching to healthier lifestyles, even for those who experience no long-term WL.

# Supplementary material

Supplementary material is available at European Journal of Preventive Cardiology.

## Acknowledgements

We thank the Dietary Intervention Randomized Controlled Trial (DIRECT), CENTRAL, and DIRECT PLUS participants for their cooperation. We thank the consultants, healthcare providers, workplace cafeteria managers, adviser researchers, and medical staff at the Nuclear Research Centre Negev (NRCN) for their invaluable contributions.

## **Author contributions**

IShai was the principal investigator of the clinical trials. I.Shai, M.J.S., D.D.W., P.K., M.B., N.K., M.S., U.C., and B.I. obtained funding. A.Y.M., G.T., H.Z., E.R., Y.G., and D.S. coordinated the clinical trials and collected the raw data. A.Y.M., G.T., H.Z., YG, A.R., M.B., U.C., M.K., P.K., D.D.W., L.Q., L.L., F.B.H., M.I.S., and I.Shai contributed to the design of the current study. A.Y.M., G.T., H.Z., E.R., and Y.G. set up the database and entered the data. A.Y.M., G.T., H.Z., E.R., Y.G., D.S., I.Shelef, and I.Shai screened and enrolled the participants, obtained informed consent, and provided patient care and documentation of study data. I.Shelef supervised the radiology scans. M.K. and P.K. isolated DNA samples and performed the DNA methylation analysis. U.C. and B.I. performed laboratory and metabolomic analysis. N.K. performed the proteomic analysis. A.Y.M., G.T., and I.Shai verified the underlying data, analyzed all findings, and take responsibility for the integrity and accuracy of the data analysis. A.Y.M., G.T., and I.Shai wrote the manuscript. All authors contributed to data interpretation, revised, and approved the manuscript. A.Y.M., G.T., and I.Shai take final responsibility for the decision to submit the study for publication.

### Funding

This study was supported by grants from the Deutsche Forschungsgemeinschaft (DFG, German Research Foundation)—Project number 209933838—Collaborative Research CenterCentre SFB1052 'Obesity Mechanisms', to I.Shai (B11), to MB (B1), NK (B4), and PK (B07). A.Y.M. was supported by the Council for Higher Education—Zuckerman support programme for outstanding postdoctoral female researchers. None of the funding providers for any of the trials were involved in any stage of the design, conduct or analysis of the study and they had no access to the study results before publication.

Conflict of interest: None declared.

## Data availability

Upon request from the corresponding author, deidentified participants' data will be available to others after publication, as will the study protocol and the statistical analysis plan. DNA methylation raw data are available in the ArrayExpress repository: tables https://www.ebi.ac.uk/biostudies/ arrayexpress/studies/E-MTAB-8956 and https://www.ebi.ac.uk/biostudies/ arrayexpress/studies/E-MTAB-12527.

# Disclaimer

M.B. received honoraria as a consultant and speaker from Amgen, Astra-Zeneca, Bayer, Boehringer-Ingelheim, Lilly, Novo Nordisk, Novartis, Pfizer, and Sanofi. All other authors declare no conflicts of interest.

## References

- Geiss LS, James C, Gregg EW, Albright A, Williamson DF, Cowie CC, et al. Diabetes risk reduction behaviors among U.S. Adults with prediabetes. Am J Prev Med 2010;38: 403–409.
- Dansinger ML, Gleason JA, Griffith JL, Selker HP, Schaefer EJ. Comparison of the atkins, ornish, weight watchers, and zone diets for weight loss and heart disease risk reduction: a randomized trial. JAMA 2005;293:43–53.
- Moore LL, Visioni AJ, Mustafa Qureshi M, Bradlee ML, Ellison RC, D'Agostino R et al. Weight loss in overweight adults and the long-term risk of hypertension: the framingham study. Arch Intern Med 2005;165:1298–1303.
- Harrington M, Gibson S, Cottrell RC. A review and meta-analysis of the effect of weight loss on all-cause mortality risk. Nutr Res Rev 2009;22:93–108.
- Papandonatos GD, Pan Q, Pajewski NM, Delahanty LM, Peter I, Erar B, et al. Genetic predisposition to weight loss and regain with lifestyle intervention: analyses from the

diabetes prevention program and the Look AHEAD randomized controlled trials. *Diabetes* 2015;**64**:4312–4321.

- Lamiquiz-Moneo I, Mateo-Gallego R, Bea AM, Dehesa-García B, Pérez-Calahorra S, Marco-Benedí V, et al. Genetic predictors of weight loss in overweight and obese subjects. Sci Rep 2019;9:1–9.
- Strohacker K, McCaffery JM, Maclean PS, Wing RR. Adaptations of leptin, ghrelin or insulin during weight loss as predictors of weight regain: a review of current literature. *Int J Obes* 2013;**38**:388–396.
- Larsen JK, Geenen R, Maas C, de Wit P, van Antwerpen T, Brand N, *et al.* Personality as a predictor of weight loss maintenance after surgery for morbid obesity. *Obes Res* 2004; 12:1828–1834.
- Bandín C, Martinez-Nicolas A, Ordovás JM, Madrid JA, Garaulet M. Circadian rhythmicity as a predictor of weight-loss effectiveness. *Int J Obes* 2013;38:1083–1088.
- Teixeira PJ, Palmeira AL, Branco TL, Martins SS, Minderico CS, Barata JT, et al. Who will lose weight? A reexamination of predictors of weight loss in women. Int J Behav Nutr Phys Act 2004;1:1–12.
- Williamson DA, Bray GA, Ryan DH. Is 5% weight loss a satisfactory criterion to define clinically significant weight loss? *Obesity* 2015;23:2319–2320.
- Ryan DH, Yockey SR. Weight loss and improvement in comorbidity: differences at 5%, 10%, 15%, and over. Curr Obes Rep 2017;6:187–194.
- Fayh APT, Lopes AL, Da Silva AMV, Reischak-Oliveira Á, Friedman R. Effects of 5% weight loss through diet or diet plus exercise on cardiovascular parameters of obese: a randomized clinical trial. *Eur J Nutr* 2013;**52**:1443–1450.
- Aslam M, Eckhauser AW, Dorminy CA, Dossett CM, Choi L, Buchowski MS. Assessing body fat changes during moderate weight loss with anthropometry and bioelectrical impedance. Obes Res Clin Pract 2009;3:209–219.
- Anderson JW, Konz EC, Frederich RC, Wood CL. Long-term weight-loss maintenance: a meta-analysis of US studies. Am J Clin Nutr 2001;74:579–584.
- Tremblay A, Fogelholm M, Jalo E, Westerterp-Plantenga MS, Adam TC, Huttunen-Lenz M, et al. What is the profile of overweight individuals who are unsuccessful responders to a low-energy diet? A PREVIEW sub-study. Front Nutr 2021;8:779.
- Aasbrenn M, Schnurr TM, Have CT, Svendstrup M, Hansen DL, Worm D, et al. Genetic determinants of weight loss after bariatric surgery. Obes Surg 2019;29:2554–2561.
- Brennan AM, Standley RA, Yi F, Carnero EA, Sparks LM, Goodpaster BH. Individual response variation in the effects of weight loss and exercise on insulin sensitivity and cardiometabolic risk in older adults. *Front Endocrinol (Lausanne)* 2020;**11**:632.
- Melanson EL, Keadle SK, Donnelly JE, Braun B, King NA. Resistance to exercise-induced weight loss: compensatory behavioral adaptations. *Med Sci Sports Exerc* 2013;45: 1600–1609.
- Franz MJ, VanWormer JJ, Crain AL, Boucher JL, Histon T, Caplan W, et al. Weight-loss outcomes: a systematic review and meta-analysis of weight-loss clinical trials with a minimum 1-year follow-up. J Am Diet Assoc 2007;107:1755–1767.
- Fogarasi A, Gonzalez K, Dalamaga M, Magkos F. The impact of the rate of weight loss on body composition and metabolism. *Curr Obes Rep* 2022;**11**:33–44.
- Blüher M, Rudich A, Klöting N, Golan R, Henkin Y, Rubin E, et al. Two patterns of adipokine and other biomarker dynamics in a long-term weight loss intervention. *Diabetes Care* 2012;**35**:342–349.
- Shai I, Schwarzfuchs D, Henkin Y, Shahar DR, Witkow S, Greenberg I, et al. Weight loss with a low-carbohydrate, Mediterranean, or low-fat diet. N Engl J Med 2008;359:229–241.
- Gepner Y, Shelef I, Schwarzfuchs D, Zelicha H, Tene L, Yaskolka Meir A, et al. Effect of distinct lifestyle interventions on mobilization of fat storage pools. *Circulation* 2018;**137**: 1143–1157.
- Yaskolka Meir A, Rinott E, Tsaban G, Zelicha H, Kaplan A, Rosen P, et al. Effect of green-Mediterranean diet on intrahepatic fat: the DIRECT plus randomised controlled trial. *Gut* 2021;**70**:2085–2095.
- Hasan B, Nayfeh T, Alzuabi M, Wang Z, Kuchkuntla AR, Prokop LJ, et al. Weight loss and Serum lipids in overweight and obese adults: a systematic review and meta-analysis. J Clin Endocrinol Metab 2020;105:3695–3703.
- Koutoukidis DA, Koshiaris C, Henry JA, Noreik M, Morris E, Manoharan I, et al. The effect of the magnitude of weight loss on non-alcoholic fatty liver disease: a systematic review and meta-analysis. *Metabolism* 2021;**115**:154455.
- Keller M, Meir AY, Bernhart SH, Gepner Y, Shelef I, Schwarzfuchs D, et al. DNA methylation signature in blood mirrors successful weight-loss during lifestyle interventions: the CENTRAL trial. Genome Med 2020;12:1–18.
- Schwarzfuchs D, Golan R, Shai I. Four-year follow-up after two-year dietary interventions. N Engl J Med 2012;367:1373–1374.
- Turkkila E, Pekkala T, Merikallio H, Merikukka M, Heikkilä L, Hukkanen J, et al. Five-year follow-up of a randomized weight loss trial on a digital health behaviour change support system. *Int J Obes (Lond)*. 2025. doi:10.1038/s41366-025-01742-4. Epub ahead of print. PMID: 40089569.

- Group A for H in D (Look AS). Association of weight loss maintenance and weight regain on 4-year changes in CVD risk factors: the action for health in diabetes (Look AHEAD) clinical trial. *Diabetes Care* 2016;**39**:1345–1355.
- Ostrominski JW, Plutzky J, Sattar N, Vaduganathan M, Aroda VR. Obesity management: a foundational cardiovascular health priority. *Eur J Prev Cardiol.* 2025:zwaf043. doi:10.1093/eurjpc/zwaf043. Epub ahead of print. PMID: 39911097.
- Tuccinardi D, Watanabe M, Masi D, Monte L, Meffe LB, Cavallari I, et al. Rethinking weight loss treatments as cardiovascular medicine in obesity, a comprehensive review. *Eur J Prev Cardiol* 2024;**31**:1260–1273.
- Lee H, Rhee T-M, Park HE, Choi SY. Association between cumulative metabolic risk exposure and cardiovascular disease: a nationwide cohort of over 3.6 million young adults. *Eur J Prev Cardiol* 2024;**31**(10):1288–1300.
- Di Angelantonio E, Kaptoge S, Wormser D, Willeit P, Butterworth AS, Bansal N, et al.Association of cardiometabolic multimorbidity with mortality. JAWA 2015;314:52–60.

- Domanski MJ, Wu CO, Tian X, Hasan AA, Ma X, Huang Y, et al. Association of incident cardiovascular disease with time course and cumulative exposure to multiple risk factors. J Am Coll Cardiol 2023;81:1151–1161.
- 37. Arnett DK, Blumenthal RS, Albert MA, Buroker AB, Goldberger ZD, Hahn EJ, et al. 2019 ACC/AHA guideline on the primary prevention of cardiovascular disease: executive summary: a report of the American College of Cardiology/American Heart Association task force on clinical practice guidelines. J Am Coll Cardiol 2019;74:1376–1414.
- Visseren FLJ, MacH F, Smulders YM, Carballo D, Koskinas KC, Bäck M, et al. 2021 ESC guidelines on cardiovascular disease prevention in clinical practice. Eur Heart J 2021;42: 3227–3337.
- Koskinas KC, Van Craenenbroeck EM, Antoniades C, Blüher M, Gorter TM, Hanssen H, et al. Obesity and cardiovascular disease: an ESC clinical consensus statement. Eur Heart J 2024;45:4063–4098.